Author's response to reviews

Title: Challenging the role of social norms regarding body weight as an explanation for weight, height, and BMI misreporting biases: Development and application of a new approach to examining misreporting and misclassification bias in surveys

Authors:

Jonathan R Brestoff (jon.brestoff@gmail.com)  
Ivan J Perry (i.perry@ucc.ie)  
Jan Van den Broeck (jvdbroeck1959@gmail.com)

Version: 8 Date: 13 May 2011

Author's response to reviews: see over
Dear Editors:

Thank you for the final proof. This version contains corrections to very minor formatting or typographical errors. With these incorporated, we find no other errors at this time. Thank you very much.

Sincerely, and on behalf of my co-authors,

Jonathan R Brestoff